Open Access Full Text Article

## ORIGINAL RESEARCH A Case–Control Study of Type 2 Diabetes Mellitus and Periodontitis in Saudi Arabian Adults

This article was published in the following Dove Press journal: Journal of Multidisciplinary Healthcare

Mir Faeq Ali Quadri Hammam I Fageeh<sup>2</sup> Wael Ibraheem<sup>2</sup> Abbas Jessani<sup>3</sup>

<sup>1</sup>Department of Preventive Dental Sciences, Division of Dental Public Health, College of Dentistry, Jazan University, Jazan, Saudi Arabia; <sup>2</sup>Department of Preventive Dental Sciences, Division of Periodontology, College of Dentistry, Jazan University, lazan, Saudi Arabia; <sup>3</sup>College of Dentistry, Schulich School of Medicine & Dentistry, Toronto, Canada

Objective: The relationship between type 2 diabetes mellitus (T2DM) and periodontitis is bidirectional and has been investigated. However, the evidence from the middle-eastern region is sparse. The current report assessed the association between uncontrolled T2DM and periodontal status from a sample of the Saudi Arabian adult population.

Methods: A case-control study was carried out. Cases were adults diagnosed with periodontitis (clinical attachment loss  $\geq 1$  mm) and controls were patients from the same dental setting with no gum conditions matched with age, sex, and location. Diabetes was recorded using HbA1c readings. The other health conditions including hypertension, epilepsy, bronchitis, thyroid disorders, and arthritis were obtained from medical records. Data on the use of tobacco and related products (smoking, khat/qat, sheesha, shammah) were gathered using a self-perceived questionnaire. Frequencies, percentages, p-values, crude and adjusted odds ratios (OR) with 95% confidence intervals were computed.

**Results:** Overall sample comprises 166 cases and 332 controls with a mean age of 37.5 years. Multivariable analysis indicated uncontrolled T2DM as an important predictor for periodontitis among Saudi Arabian adults, and they had nearly three times greater odds (OR: 2.779; 95% CI: 1.425–5.419; p=0.003) of being diagnosed with periodontitis in contrast to non-diabetics. Secondary findings revealed that cigarette ever-users were two times more likely to be suffering from periodontitis than never-users, and those brushing once per day or less had five times greater odds of developing periodontitis as compared to those brushing twice daily.

**Conclusion:** To conclude, the current evidence from Saudi Arabia is supportive of earlier studies and an awareness of this association is warranted among all healthcare providers and patients in the region for early detection of periodontitis.

Keywords: type-2 diabetes mellitus, adult periodontitis, case-control study, association, Saudi Arabia

#### Plain Language Summary

The current study is the first to gather evidence from Jazan region of Saudi Arabia to demonstrate the relationship between T2DM and periodontitis. Findings reveal that the individuals with uncontrolled type-2 diabetes mellitus (T2DM) are more likely to develop periodontitis in comparison to the non-diabetic population. As the relationship between diabetes and periodontitis is bidirectional, selection of appropriate study design while establishing the relationship is essential. The idea is to address temporality by demonstrating that diabetes precedes periodontitis. The HbA1c level indicates uncontrolled blood sugar level for the past few months and one could assume that the people with higher than normal HbA1c level may have been suffering from uncontrolled T2DM for many months. However, the current study do not specify the likely number of months that are needed to develop periodontitis for an individual suffering from uncontrolled T2DM and also the plausible

Correspondence: Mir Faeq Ali Quadri Department of Preventive Dental Sciences, Division of Dental Public Health, College of Dentistry, Jazan University, Jazan City 45142, Saudi Arabia Tel +00966 5989 59409 Email dr.faeq.quadri@gmail.com



© 2020 Quadri et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/rerms. bp and incorporate the Creative Commons Attribution — Non Commercial (unported, v3.0) License (http://creativecommons.org/licenses/by-nc/3.0/). By accessing the work you hereby accept the Terms. Non-commercial uses of the work are permitted without any further permission for Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, please see paragraphs 4.2 and 5 of our Terms (https://www.dovepress.com/terms.php).

mechanism. An awareness of this association among healthcare providers and patients is essential as it may contribute towards early detection and prevention of periodontitis among the diabetics.

## Introduction

Periodontitis is one of the most prevalent oral diseases in the adult population worldwide.<sup>1,2</sup> A common clinical sign of periodontitis is loss of dental connective tissue and bone, caused by inflammatory host response secondary to bacterial infection.<sup>3</sup> There are certain pathophysiological conditions that have the ability to disrupt the symbiotic equilibrium of microbes in the oral cavity, ultimately leading to periodontitis.<sup>4–6</sup> One such debilitating physiologic condition is diabetes, which is shown to be associated with periodontitis.<sup>7</sup> Among the two broad categories of diabetes, epidemiological studies have classified type 2 diabetes mellitus (T2DM) as a widespread condition, with current reports indicating approximately 285 million people with this disease, and estimating an increase of 50% by the year 2030.<sup>8</sup> Reports indicate that blood sugar levels caused by diabetes may trigger a bacterial imbalance leading to a cascade of inflammatory reactions in the periodontal apparatus, causing loss of teeth, decreased chewing ability, nutritional imbalance, and eventually, a reduced oral health-related quality of life.9,10

The certainty of association is agreed upon by many experts, and systematic review and meta-analysis have been performed to demonstrate and evaluate the effect of diabetes on periodontitis.<sup>11</sup> There are findings that report otherwise. For instance, a systematic review had stated that patients with diabetes had the same extent of periodontal disease compared with non-diabetics.<sup>12</sup> The contrast in findings is largely attributed to the quality of methodology adopted by the included studies in the reviews. A recent systematic review had revealed an increased heterogeneity ( $I^2 = 89.7\%$ , p < 0.001) among the included reports especially in terms of their study design.<sup>13</sup> In line with this, another systematic review that demonstrated a significant association between diabetes and periodontitis showed that out of the 49 cross-sectional and 8 longitudinal studies about 26 could have demonstrated better and precise findings if the assessment of variables was standardized, sample recruitment process was robust, and if the findings were adjusted for confounding variables.<sup>14</sup>

The Center for Evidence-Based Medicine (CEBM) specifies that proper study design and methodology and judgmental intuition of other potential factors are highly

important to estimate an effect between the exposure and the outcome.<sup>15</sup> Till date, there has been no evidence provided from an ethnically similar population of Saudi Arabian adults to substantiate the association of uncontrolled diabetes and periodontitis by utilizing the gold standard glycated hemoglobin (HbA1c) test reports and clinical attachment loss, respectively. It is proclaimed that epidemiological studies are appropriate to investigate the association; among them, cross-sectional studies exhibit limited causal inference as the exposure and outcomes are examined concurrently.<sup>16</sup> Cohort studies provide better evidence but are not always feasible due to the huge enrollment required to demonstrate an effect, and the likely loss of follow-up data for some participants.<sup>17</sup> Due to these reasons, it is put forth that a case-control study design satisfies the criteria of temporality and is a more feasible approach in determining the association. With this, the current study aimed to investigate the association between uncontrolled T2DM (exposure) and periodontitis (outcome) among Saudi Arabian adult population; and it is hypothesized that the odds of developing periodontitis in participants with uncontrolled T2DM are higher than in their non-diabetic counterparts.

The rationale for this investigation is the plausible influence of pathogenic mechanism in the individuals who have uncontrolled diabetes in comparison to non-diabetic,<sup>18,19</sup> and the data from the Saudi Arabian population are limited.

#### **Methods**

## Study Design and Study Location

A hospital-based case–control study was designed following the STROBE (Strengthening the Reporting of Observational Studies in Epidemiology) guidelines.<sup>20,21</sup> The current study was conducted in the Jazan province of Saudi Arabia, which is located in the south-western part of the Arabian Peninsula, bordering Yemen. The site of data collection was the Dental Center affiliated with Jazan University. It is the largest oral health service provider in the region, catering to the needs of more than 1.5 million citizens, with the complete oral health cost borne by the Government of Saudi Arabia.

#### Sample Size Estimation

In a previous nationwide study, the age-specific prevalence of T2DM among Saudi Arabian residents was reported to be 40.6%, with the Jazan region contributing to around 9.4% of the cases.<sup>22</sup> Due to lack of prevalence estimates of periodontitis directly from this region, the sample size calculation was based on data from the most recent available study performed in another region of Saudi Arabia.<sup>23</sup> This included 80% power, 95% precision, and an assumption of 0.5% of controls being exposed to the risk factor. The sample, was inclusive of two controls for each case (1:2). The estimated sample size required to reject the null hypothesis, with a 10% addition for dropouts was determined to be 66 cases and 132 controls. However, to attain a better precision all the available cases (n = 166) and twice (n = 332) the number of controls in the months from August to October of 2018 were recruited.

## Selection Criteria

Saudi Arabian adults confirming their identity by displaying the national card were recruited as cases and controls for the current study. This was done to ensure similar ethnic backgrounds and cultural habits of study participants. Patients presenting with physical and mental disabilities, or patients who did not provide signed consent, were excluded.

## Defining the Cases and Controls

Cases were defined as clinically diagnosed periodontitis patients who underwent oral health examinations at the Dental Center affiliated with Jazan University by two calibrated specialists using the method prescribed by the American Academy of Periodontology (2014).<sup>24</sup> Controls were adults selected from the same Dental Center, who had not been diagnosed with periodontitis, and adjusted for age ( $\pm$ 5 years), sex, and location.

## Assessment of Outcome Variable

Assessment of outcome was conducted using the standardized classification system for periodontal diseases and conditions.<sup>24,25</sup> A Williams' periodontal probe (Hu-Friedy<sup>®</sup> PW6) was used to measure clinical parameters, including clinical attachment loss (CAL) and probing depths (PD). Six sites per tooth were assessed, and a diagnosis of periodontitis was made when subjects presented with at least two sites with a PD  $\geq$  4 mm and a CAL  $\geq$  1 mm (one on each tooth).<sup>25</sup>

## Assessment of Exposure Variables

Medical reports were checked to confirm the history of T2DM, based on the date of first diagnosis confirmed through glycated hemoglobin (HbA1c) tests. Then,

patients with the current HbA1c level higher than HbA1c  $\geq$ 7%;  $\geq$ 53 mmol/mol were defined as those having uncontrolled diabetes.<sup>26-28</sup> A medical history of hypertension, epilepsy, bronchitis, thyroid disorders, and arthritis was confirmed from the participant's medical records, and responses were dichotomized (yes/no). Oral hygiene practice was recorded by asking the participants if they preferred brushing their teeth twice a day, once a day, or less than once a day. The response received was dichotomized, and "twice per day" was considered as a good oral hygiene practice, while the other options were considered as a fair/ poor oral hygiene practice. Data on tobacco and related products (smoking, khat/qat, sheesha, and shammah) commonly used in the region were gathered through a patientevaluation questionnaire, and responses were subsequently dichotomized to classify the patients as ever-users and never-users. Previous users of the mentioned products who had abstained for more than 5 years were also considered as never-users.

## Specialist Calibration

Periodontitis among the study participants was diagnosed independently by two American board-certified periodontists who were part of the current study. The first 10% of the recruited cases were cross-examined to assess the calibration. The inter-examiner reliability was calculated as Kappa (unweighted Kappa), and the result obtained ranged between 0.75 and 0.90 with an average of 0.84  $\pm$  0.03 for the eight patients, achieving a satisfactory calibration agreement.<sup>29</sup>

## Data Collection Process

The data collection process involved a clinical examination of the suspected cases and a total of 166 cases were recruited. Twice the number of controls (n = 332) were recruited from other departments of the same dental setting. Both cases and controls were subsequently assessed for uncontrolled T2DM and other exposure variables using the parameters mentioned earlier. History of periodontitis and diabetes was verified using the available electronic health records of the patients to observe that the outcome followed the exposure.

## Statistical Analyses

Relationship between periodontitis and T2DM, as well as other covariates, were first identified using the Chi-square or Fisher's exact test, as appropriate. The associations were explored further with binary and multiple logistic regression analyses, after adjusting for confounders. Data were presented as frequencies, percentages, p-values, crude OR and adjusted OR; with their 95% CI. Data entry and statistical analyses were performed using the Statistical Package for the Social Sciences v.24 (SPSS, IBM Statistics). A p-value of <0.05 was considered as significant.

## Results

The sample consisted of 166 cases and 332 controls (1:2) from the available pool of patients. The mean age of the study sample was 37.5 years (SD = 5.1). None were mentally or physically challenged or had gestational diabetes, renal disorders, HIV-infection, cancers, or kidney disorders. Estimates for hypotension, T1DM, pregnancy, epilepsy, thyroid, and shammah (smokeless tobacco, could not be computed as these conditions were only present in either the cases or controls, and were even excluded from the multivariable models.

Findings from the relationship of periodontitis with general health conditions are reported in Table 1. T2DM, hypertension, diabetes, epilepsy, allergic conditions, asthma, and bronchitis were the reported general health conditions in the study sample. Among them, T2DM was more common in the cases and the controls. A significant relationship of T2DM and hypertension with periodontitis was observed. Hypertension was reported in 4.8% of the cases and in 0.6% of the controls (p = 0.003), while T2DM was reported in 28.9% of the cases and 5.7% of the controls (p < 0.001) (Table 1).

Findings from the relationship of periodontitis with oral hygiene habits and substance use are reported in Table 2. Brushing once per day or less as opposed to twice daily was observed significantly more in cases (39.1%) compared to controls (11.4%) (p < 0.001). Smoking (p = 0.006) and khat/qat (p = 0.002) use was also found to be significantly higher in patients with periodontitis (Table 2).

The findings from the binary and multiple logistic regression analyses are presented in Table 3. The crude analysis identified T2DM and hypertension as significantly associated with periodontitis. Patients with T2DM and hypertension had six times (OR: 6.701; 95% CI: 3.783—11.872) and eight times (OR: 8.354; 95% CI: 2–39.799) higher odds, respectively, of being diagnosed with periodontitis. The multivariate analyses were adjusted for asthma, bronchitis, and sheesha use. Adjusted analysis revealed that T2DM patients had nearly three times greater odds (OR: 2.779; 95% CI: 1.425–5.419; p = 0.003) of being diagnosed with periodontitis than their non-T2DM counterparts (Table 3).

Cigarette smoking, khat use, and oral hygiene practice were significantly associated with periodontitis in the crude analysis. The adjusted analysis however revealed that patients with the habit of cigarette smoking were two times (OR: 2.088; 95% CI: 1.013–4.305; p = 0.046) more likely to have periodontitis. Khat use was not significantly related. Finally, patients brushing only once per day or less had a five times greater chance of developing periodontitis as compared to patients brushing twice daily (Table 3).

#### Discussion

Epidemiologic studies in Arab nations, including Saudi Arabia suggest a consistent rise in the T2DM cases that need immediate attention.<sup>22,30</sup> The Center for Disease Control (CDC) has reported that individuals diagnosed with diabetes are more likely to suffer from severe inflammation

 Table I General Health Conditions and Their Relationship with Periodontitis (N= 498)

General Health Conditions	Cases (N=166) n (%)	Controls (N=332) n (%)	Total n (%)	p-value	
T2DM	48(28.9)	19(5.7)	67(13.4)	<0.001*	
TIDM	l (0.6)	0(0)	I (0.2)	0.333†	
Hypertension	8(4.8)	2(0.6)	10(2.0)	0.003*†	
Hypotension	0(0)	2(0.6)	2(0.4)	0.555 <sup>†</sup>	
Asthma	5(3.0)	13(3.9)	18(3.6)	0.611	
Pregnancy	0(0)	2(0.6)	2(0.4)	0.555 <sup>†</sup>	
Epilepsy	l (0.6)	0(0)	I (0.2)	0.333†	
Arthritis	3(1.8)	5(2.7)	8(1.6)	1.000 <sup>†</sup>	
Allergies	l (0.6)	9(2.7)	10(2)	0.176 <sup>†</sup>	
Thyroid disorders	l (0.6)	0(0)	I (0.2)	0.333†	
Bronchitis	l (0.6)	3(0.9)	4(0.8)	1.000 <sup>†</sup>	

Notes: \*Statistically significant (p<0.05), † Fishers' test.

Habits	its Cases (N=166) n (%) Controls (N=332) n (%)		Total n (%)	p-value
Cigarette <sup>a</sup>	27(16.2)	27(8.1)	54(10.8)	0.006*
Khat/qat <sup>a</sup>	20(12.0)	15(4.5)	35(7.0)	0.002*
Sheesha <sup>a</sup>	l (0.6)	I (0.3)	2(0.4)	1.000 <sup>†</sup>
Shammah <sup>a</sup>	l (0.6)	0(0)	I (0.2)	0.333 <sup>†</sup>
Brushing once a day or less <sup>\$</sup>	65(39.1)	38(11.4)	103(20.6)	<0.001*

Table 2 Relationship of Oral Hygiene Practice and Substance Use with Periodontitis (N= 498)

Notes: \*Statistically significant (p<0.05), †Fishers' test, <sup>\$</sup>Reference category: brushing twice a day, <sup>a</sup>Ever-users vs never-users.

Table 3 Binary and Multivariate Logistic Regression Analysis Demonstrating the Effect Various Exposure Variables on PeriodontalStatus (N= 498)

Risk Factors	Crude	Crude			Adjusted <sup>†</sup>		
	OR	95% CI	p-value	OR	95% CI	p-value	
T2DM	6.701	(3.783–11.872)	<0.001*	2.779	(1.425–5.419)	0.003*	
Hypertension	8.354	(2-39.799)	0.003*	3.279	(0.507–21.191)	0.212	
Arthritis	1.204	(0.284–5.099)	1.000	0.459	(0.095-2.230)	0.334	
Allergies	0.218	(0.027–1.731)	0.176	0.268	(0.029–2.474)	0.245	
Cigarette	2.194	(1.241-3.880)	0.006*	2.088	(1.013-4.305)	0.046*	
Khat/qat	2.895	(1.441–5.816)	0.002*	2.206	(0.922-5.279)	0.075	
Brushing once a day <sup>\$</sup>	4.979	(3.144–7.885)	0.000*	5.630	(3.514–9.021)	<0.001*	

**Notes:** \*Statistically significant (p<0.05); †Adjusted for asthma, bronchitis, and sheesha; <sup>\$</sup>Reference category: brushing twice a day. **Abbreviation:** OR, odds ratio.

and inadequate white blood cell functions, which can subsequently lead to destruction of periodontal tissues.<sup>31</sup> However, to properly study the effect of diabetes on periodontitis, it was important to distinguish diabetic patients with uncontrolled blood sugar levels from those with controlled blood sugar levels. This is because the degree of glycemic control may determine the vulnerability of the immune defense mechanism, which ultimately leads to the damage of periodontal tissues. Further, there are reports that suggest that T2DM, in particular, has the ability to remain undiagnosed for a long time as the affected patients retain their capacity to produce insulin, and the symptoms are usually missed.<sup>32</sup> Thus, gold-standard measures as opposed to subjective measures were required to confirm the glycemic control in the diabetic patients.

Although the relationship between diabetes mellitus and periodontitis is well studied, the current study is the first in Saudi Arabia to investigate the effect of uncontrolled T2DM on periodontal status using a case–control study design. After adjusting for other variables like systemic diseases, substance used (Khat, sheesha, and shammah), and socio-demographic characteristics (age, sex, location, and ethnicity), it is observed that the subjects with uncontrolled diabetes (T2DM) had nearly three times greater likelihood of developing periodontitis. A recent research in Saudi Arabia with a different study design aiming to compare the levels of Advance Glycation End-products in the gingival crevicular fluids of patients suffering from chronic periodontitis, with and without T2DM, supported the findings of the current study.<sup>33</sup> However, the stated limitation was the inclusion of patients with controlled diabetes mellitus along with uncontrolled diabetes mellitus. In addition, one group of patients with uncontrolled diabetes was prescribed with medications and were still reported as having abnormal glycemic control, indicating that other risk factors or covariates that could have led to this uncontrolled state of blood sugar level were not considered.<sup>33</sup>

A longitudinal study among the Gullah African Americans also showed that uncontrolled T2DM relates to a nearly two times higher magnitude of periodontal tissue destruction compared to their counterparts (controlled diabetics and non-diabetics).<sup>34</sup> The results are further substantiated by other investigations from the United Kingdom,<sup>35</sup> United States of America,<sup>36</sup> India,<sup>37</sup> and Spain.<sup>38</sup> A systematic review and meta-regression of longitudinal studies also confirms this association. However, these findings from meta-regression indicated a 25% variability between the published studies with respect to their sample size.<sup>13</sup> It further reiterated that some of the included studies did not have a representative sample,<sup>13</sup> and evidence suggest that a small sample size often overestimates the association between the exposure and the outcome.<sup>39</sup>

Secondary findings from the current study revealed that the patients who brushed their teeth once per day or less had a five times greater chance of developing periodontitis than patients who brushed twice per day. Affirming this, Han et al (2017) suggested that maintaining a good personal oral hygiene through brushing, flossing, and professional oral prophylactic treatment is an important strategy to prevent gingival diseases, including periodontitis.<sup>40</sup> Similarly, other studies have also unequivocally stated that poor oral hygiene practices increase the risk of periodontitis by nearly fivefold, and that the risk substantially reduces with regular tooth brushing and dental visits.<sup>41</sup>

The multivariable model in the current study also indicated that cigarette ever-users have nearly double the likelihood of being diagnosed with periodontitis than neverusers. This association between cigarette smoking and periodontitis is well established.<sup>42,43</sup> However, the use of khat (*Catha edulis*), which is reported to be responsible for periodontal diseases among a section of Arab populations living in the Southern border of the Arabian peninsula and Yemen,<sup>44</sup> did not demonstrate a significant relationship with the incidence of periodontitis in the current study. We suggest that there may have been a collinear effect of cigarette use and khat (*Catha edulis*) use. In support of our finding, the latest systematic review conducted by Al-Maweri et al in 2018 also reported that the available evidence suggesting the effect of khat on periodontal tissues is weak.<sup>45</sup>

The strength of the current study lies in the use of goldstandard measures for the exposure (T2DM) and the outcome (periodontitis) variables. The cases and controls from the sample of Arab adults were further matched with important covariates such as, age, sex, and location. The current report was also unique in relation to some of the earlier studies as it specifically considered patients with uncontrolled T2DM, as recommended by experts.<sup>46</sup> The data collection process of this study focused towards temporality by confirming that the outcome among patients followed their exposure to risk factors. The study also prevented selection bias by recruiting all the eligible cases. Lastly, multivariable analysis with cluster effect of common risk factors was computed to obtain the findings after taking potential confounders into account.

However, the results should be interpreted carefully, as a case–control study is not the best approach to demonstrate

a cause and effect relationship. While we acknowledge that a prospective cohort study design would be preferable for the demonstration of causal association, this was not advisable in the present scenario due to the lack of a sufficiently large sample of uncontrolled T2DM patients. Also, in a prospective study, a concern of attrition bias could have arisen when the two groups (with and without uncontrolled T2DM) had taken physician consultations for their T2DM condition. Thereby, leading to a controlled state of blood sugar level. In addition, the findings of the current study may not be completely generalizable as the study was based on data gathered from a single hospital. However, the study was performed at the largest Dental Center in the region that caters free oral health care to the residents.

Despite the limitations, the current study gives directions to the future research and provides evidence of association from a population that has not been studied earlier. Thus, the findings from this study endorse routine periodontal assessment for patients with diabetes in Saudi Arabia, especially those with a history of uncontrolled blood sugar levels. Additionally, as patients with T2DM are often not aware of their medical condition, oral healthcare providers are also advised to refer patients with diagnosed generalized periodontitis and without self-reported diabetes to general practitioners. In future, more evidences of such relationship from the Saudi Arabian population is needed. Subsequently, the definitive relationship between oral and general health should be enhanced among the researchers and clinicians across Saudi Arabia by exploring and presenting the associations of oral health problems with other conditions such as the cardiovascular disease.<sup>5</sup>

## Conclusion

To conclude, the current case–control study is the first from the Jazan region of Saudi Arabia to indicate that uncontrolled T2DM denoted by increased HbA1c level is associated with greater odds of periodontitis. An awareness of this association among the healthcare providers and patients in this region will contribute towards the overall prevention of periodontitis in Saudi Arabia.

## Abbreviations

T2DM, type-2 diabetes mellitus; HbA1C, hemoglobin A1C; CAL, clinical attachment loss; CDC, Center for Disease control; CEBM, Center for Evidence-Based Medicine.

## **Data Sharing Statement**

The dataset of the current study will be made available upon request.

# Ethics Approval and Consent to Participate

This study was approved by the Research Unit (RU), College of Dentistry, Jazan University. Written and signed consent was obtained from each participant recruited in the study. The study abides by the Helsinki Declaration presented by the World Medical Association.

## Acknowledgments

The authors would like to acknowledge the study participants. We would also like to thank the Dean, Dr Adbulwahab Alamir and the research unit at the College of Dentistry for their constant support during the current project. All authors gave their final approval and agree to be accountable for all aspects of the work. There is no conflict of interest to be declared and the study was not funded by any external source.

## **Author Contributions**

All authors made substantial contributions to conception and design, acquisition of data, or analysis and interpretation of data; took part in drafting the article or revising it critically for important intellectual content; agreed to submit to the current journal; gave final approval of the version to be published; and agree to be accountable for all aspects of the work.

## Funding

The current study is self-funded and we declare that no external funds from any organization/s has been received for performing the study or for the dissemination of the findings.

## Disclosure

The authors do not have any competing interests to report with regards to the current research.

## References

- Albandar JM. Epidemiology and risk factors of periodontal diseases. *Dent Clin North Am.* 2005;49(3):517–vi. doi:10.1016/j.cden.2005. 03.003
- 2. Dye BA. Global periodontal disease epidemiology. *Periodontol.* 2012;58(1):10–25. doi:10.1111/j.1600-0757.2011.00413.x
- Löe H. Periodontal Disease: the sixth complication of diabetes mellitus. *Diabetes Care*. 1993;16(1):329–334. doi:10.2337/diacare. 16.1.329
- Kilian M, Chapple IL, Hannig M, et al. The oral microbiome an update for oral healthcare professionals. *Br Dent J.* 2016;221 (10):657–666. doi:10.1038/sj.bdj.2016.865

- Isola G, Alibrandi A, Currò M, et al. Evaluation of salivary and serum ADMA levels in patients with periodontal and cardiovascular disease as subclinical marker of cardiovascular risk. *J Periodontol.* 2020.
- Isola G, Polizzi A, Iorio-Siciliano V, Alibrandi A, Ramaglia L, Leonardi R. Effectiveness of a nutraceutical agent in the non-surgical periodontal therapy: a randomized, controlled clinical trial. *Clin Oral Investig.* 2020. doi:10.1007/s00784-020-03397-z
- Kampoo K, Teanpaisan R, Ledder RG, McBain AJ. Oral bacterial communities in individuals with type 2 diabetes who live in southern Thailand. *Appl Environ Microbiol*. 2014;80(2):662–671. doi:10.1128/ AEM.02821-13
- RS MN L, Fernandes JK, Hermayer K. Oral health and Type 2 Diabetes. *Am J Med Sci.* 2013;345(4):271–273.
- Haag DG, Peres KG, Balasubramanian M, Brennan DS, Conditions O. Health-Related Quality of Life: A Systematic Review. J Dent Res. 2017;96(8):864–874. doi:10.1177/0022034517709737
- Isola G, Polizzi A, Patini R, Ferlito S, Alibrandi A, Palazzo G. Association among serum and salivary A. actinomycetemcomitans specific immunoglobulin antibodies and periodontitis. *BMC Oral Health.* 2020;20(1):283. doi:10.1186/s12903-020-01258-5
- Baeza M, Morales A, Cisterna C, et al. Effect of periodontal treatment in patients with periodontitis and diabetes: systematic review and meta-analysis. *J Appl Oral Sci.* 2020;28:e20190248. doi:10.1590/ 1678-7757-2019-0248
- Khader YS, Dauod AS, El-Qaderi SS, Alkafajei A, Batayha WQ. Periodontal status of diabetics compared with nondiabetics: a meta-analysis. J Diabetes Complications. 2006;20(1):59–68. doi:10.1016/j.jdiacomp.2005.05.006
- Nascimento GG, Leite FRM, Vestergaard P, Scheutz F, López R. Does diabetes increase the risk of periodontitis? A systematic review and meta-regression analysis of longitudinal prospective studies. *Acta Diabetol.* 2018;55(7):653–667. doi:10.1007/s00592-018-1120-4
- Chávarry NG, Vettore MV, Sansone C, Sheiham A. The relationship between diabetes mellitus and destructive periodontal disease: a meta-analysis. Oral Health Prev Dent. 2009;7(2):107–127.
- CEBM. Center for Evidence Based Medicine; Study Designs. Nuffield Department of Primary Care and Health Sciences; 2019.
- Checkoway H, Pearce N, Kriebel D. Selecting appropriate study designs to address specific research questions in occupational epidemiology. *Occup Environ Med.* 2007;64(9):633–638. doi:10.1136/oem.2006.029967
- Sedgwick P. Bias in observational study designs: prospective cohort studies. *BMJ*. 2014;349(dec19 2):g7731. doi:10.1136/bmj.g7731
- Lalla E, Cheng B, Lal S, et al. Diabetes-related parameters and periodontal conditions in children. *J Periodontal Res.* 2007;42 (4):345–349. doi:10.1111/j.1600-0765.2006.00955.x
- Taylor JJ, Preshaw PM, Lalla E. A review of the evidence for pathogenic mechanisms that may link periodontitis and diabetes. *J Clin Periodontol.* 2013;40(Suppl 14):S113. doi:10.1111/jcpe.12059
- Malta M, Cardoso LO, Bastos FI, Magnanini MMF, Silva CMFPD. Iniciativa STROBE: subsídios para a comunicação de estudos observacionais. *Rev Saude Publica*. 2010;44(3):559–565. doi:10.1590/S0034-89102010000300021
- 21. von Elm E, Altman DG, Egger M, et al. The Strengthening the Reporting of Observational Studies in Epidemiology (STROBE) Statement: guidelines for reporting observational studies. *Int J Surg.* 2014;12(12):1495–1499. doi:10.1016/j.ijsu.2014.07.013
- Al-Rubeaan KA-M, Al-Manaa H, Khoja T, et al. M; and Al Ghamdi, A. The Saudi Abnormal Glucose Metabolism and Diabetes Impact Study (SAUDI-DM). *Ann Saudi Med.* 2014;34(6):465–475. doi:10.5144/0256-4947.2014.465
- 23. Al-Sharabi AK, Shuga-Aldin H, Ghandour I, Al-Hebshi NN. Qat chewing as an independent risk factor for periodontitis: a cross-sectional study. *International Journal of Dentistry*. 2013;2013:317640. doi:10.1155/2013/317640

- Das SS. American Academy of Periodontology Task Force report on the update to the 1999 classification of periodontal diseases and conditions. *J Periodontol*. 2015;86(7):835–838. doi:10.1902/ jop.2015.157001
- 25. Armitage GC. Development of a classification system for periodontal diseases and conditions. *Northwest Dent.* 2000;79(6):31–35.
- 26. Aschner P. New IDF clinical practice recommendations for managing type 2 diabetes in primary care. *Diabetes Res Clin Pract.* 2017;132:169–170. doi:10.1016/j.diabres.2017.09.002
- The American Diabetes Association (ADA).
   Classification and Diagnosis of Diabetes: standards of Medical Care in Diabetes—2018. *Diabetes Care*. 2018;41(Suppl Supplement 1):S13–S27. doi:10.2337/dc18-S002
- Borgharkar SS, Das SS. Real-world evidence of glycemic control among patients with type 2 diabetes mellitus in India: the TIGHT study. *BMJ Open Diabetes Res Care.* 2019;7(1):e000654. doi:10.1136/bmjdrc-2019-000654
- Landis JR, Koch GG. The measurement of observer agreement for categorical data. *Biometrics*. 1977;33(1):159–174. doi:10.2307/ 2529310
- 30. Alotaibi A, Perry L, Gholizadeh L, Al-Ganmi A. Incidence and prevalence rates of diabetes mellitus in Saudi Arabia: an overview. *J Epidemiol Glob Health*. 2017;7(4):211–218. doi:10.1016/j. jegh.2017.10.001
- CDC. Centre for Disease Control and Prevention; National Health Interview Survey (NHIS)-2007; 2018.
- DeFronzo RA. Insulin resistance: a multifaceted syndrome responsible for NIDDM, obesity, hypertension, dyslipidaemia and atherosclerosis. *Neth J Med.* 1997;50(5):191–197. doi:10.1016/ S0300-2977(97)00012-0
- Akram Z, Alqahtani F, Alqahtani M, Al-Kheraif AA, Javed F. Levels of advanced glycation end products in gingival crevicular fluid of chronic periodontitis patients with and without type-2 diabetes mellitus. *J Periodontology*. 2020;91(3):396–402. doi:10.1002/ JPER.19-0209
- 34. Bandyopadhyay D, Marlow NM, Fernandes JK, Leite RS. Periodontal disease progression and glycaemic control among Gullah African Americans with type-2 diabetes. J Clin Periodontol. 2010;37(6):501–509. doi:10.1111/j.1600-051X.2010.01564.x
- 35. Heidari E, Andiappan M, Banerjee A, Newton JT. The oral health of individuals with dental phobia: a multivariate analysis of the Adult Dental Health Survey, 2009. Br Dent J. 2017;222(8):595–604. doi:10.1038/sj.bdj.2017.361

- 36. Tsai C, Hayes C, Taylor GW. Glycemic control of type 2 diabetes and severe periodontal disease in the US adult population. *Community Dentistry Oral Epidemiology*. 2002;30(3):182–192. doi:10.1034/ j.1600-0528.2002.300304.x
- 37. Khanuja PK, Narula SC, Rajput R, Sharma RK, Tewari S. Association of periodontal disease with glycemic control in patients with type 2 diabetes in Indian population. *Front Med.* 2018;55 (7):110–119. doi:10.1007/s00592-018-1120-4
- Tapias Ledesma M. Relación entre diabetes y enfermedad periodontal en una población adulta española. *Aten Primaria*. 2016;48 (8):558–559. doi:10.1016/j.aprim.2015.11.007
- 39. Zhang Z, Xu X, Ni H. Small studies may overestimate the effect sizes in critical care meta-analyses: a meta-epidemiological study. *Crit Care*. 2010;44(3):R2. doi:10.1590/S0034-89102010000300021
- 40. Han K, Park JB. Association between oral health behavior and periodontal disease among Korean adults: the Korea National Health and Nutrition Examination Survey. *Medicine*. 2017;96(7): e6176. doi:10.1097/MD.00000000006176
- Lertpimonchai A, Rattanasiri S, Arj-Ong Vallibhakara S, Attia J, Thakkinstian A. The association between oral hygiene and periodontitis: a systematic review and meta-analysis. *Int Dent J.* 2017;67 (6):332–343. doi:10.1111/idj.12317
- Chrysanthakopoulos NA. Risk factors for the progression of periodontal disease in a Greek adult population. J Investig Clin Dent. 2017;8(2). doi:10.1111/jicd.12199
- Bhat M, Roberts-Thomson K, Do LG. Clustering of risk indicators for periodontal disease: a population-based study. *Community Dent Health.* 2015;32(3):158–162.
- 44. Al-Kholani AI. Influence of khat chewing on periodontal tissues and oral hygiene status among Yemenis. *Dent Res J (Isfahan)*. 2010;7 (1):1–6.
- Al-Maweri SA, Warnakulasuriya S, Khat SA. (Catha edulis) and its oral health effects: an updated review. *J Investig Clin Dent*. 2018;9 (1). doi:10.1111/jicd.12288
- 46. Casanova L, Hughes FJ, Preshaw PM. Diabetes and periodontal disease: a two-way relationship. *Br Dent J.* 2014;217(8):433–437. doi:10.1038/sj.bdj.2014.907

#### Journal of Multidisciplinary Healthcare

Publish your work in this journal

The Journal of Multidisciplinary Healthcare is an international, peerreviewed open-access journal that aims to represent and publish research in healthcare areas delivered by practitioners of different disciplines. This includes studies and reviews conducted by multidisciplinary teams as well as research which evaluates the results or conduct of such teams or healthcare processes in general. The journal

Submit your manuscript here: https://www.dovepress.com/journal-of-inflammation-research-journal

#### **Dove**press

covers a very wide range of areas and welcomes submissions from practitioners at all levels, from all over the world. The manuscript management system is completely online and includes a very quick and fair peer-review system. Visit http://www.dovepress.com/testimonials. php to read real quotes from published authors.